Author's response to reviews

Title: Primary osteosarcoma of the urinary bladder following cyclophosphamide therapy for systemic lupus erythematosus: a case report

Authors:

Dilek Ertoy Baydar (dertoy@hacettepe.edu.tr)
Cigdem Himmetoglu (cigdemh@hacettepe.edu.tr)
Sertac Yazici (msertacyazici@yahoo.com)
Halil Kiziloz (halilkiziloz@gmail.com)
Haluk Ozen (hozen@hacettepe.edu.tr)

Version: 2 Date: 4 September 2008

Author's response to reviews:

A consent section stating that a written consent was obtained from the patient for publication of the manuscript and figures, has been added to the end of the manuscript.